Author’s response to reviews

Title: Assessment of two complementary influenza surveillance systems: Sentinel primary care Influenza-like Illness versus severe hospitalized laboratory-confirmed influenza using the moving epidemic method

Authors:

Nuria Torner (nuria.torner@gencat.cat)
Luca Basile (chags@gencat.cat)
Ana Martinez (a.martinez@gencat.cat)
Cristina Rius (crius@aspb.cat)
Pere Godoy (pere.godoy@gencat.cat)
Mireia Jane (mireia.jane@gencat.cat)
Angela Dominguez (angela.dominguez@ub.edu)

Version: 4 Date: 28 Jul 2019

Author’s response to reviews:

Reviewer reports:

I'm sorry , update of reference 1 has now been done


Reference has been replaced

Text has been revised for typo errors and English

Assistant Editor Comments:

1. Ethics:A reference that states that this is deemed unnecessary according to national regulations
2. Consent:The reason for verbal consent from each attending clinician has been stated
3. Availability of data: Request to nuria.torner@gencat.cat or chagas@gencat.cat